Author's response to reviews

Title: Pleural Fluid Soluble Triggering Receptor Expressed on Myeloid Cells-1 as a Marker of bacterial infection: a Meta-analysis.

Authors:

Hanssa Summah (hanssa_04@hotmail.com)
Li-Li Tao (ivydoctao@gmail.com)
Ying-Gang Zhu (robinzyq-zhu@vip.sina.com)
Hong-Ni Jiang (jhn2001@sina.com)
Jie-Ming Qu (jmqu64@yahoo.com.cn)

Version: 3 Date: 31 August 2011

Author's response to reviews: see over
Dear Editor,

We appreciate the fact that your journal is considering our manuscript for publication. We have revised the manuscript once again and changes have been made to format the manuscript according to the journal style. There have been no further changes requested by the reviewers, and thus, we did not proceed to any other changes. Thanking you in anticipation,

Kind regards,
Dr Hanssa Summah,
Dr Li-Li Tao,
Dr Ying-Gang Zhu,
Dr Hong-Ni Jiang,
Prof. Jie-ming Qu.

Corresponding author: Jie-Ming Qu. Department of Pulmonary Medicine, Huadong Hospital, Shanghai Medical College, Fudan University, Shanghai 200040, People’s Republic of China. E-mail: jmqu64@yahoo.com.cn
Tel: 0086-21-6248 3180-63504. Fax: 0086-21-62484981

Dr Hanssa Summah: Department of Pulmonary Medicine, Huadong Hospital, Shanghai Medical College, Fudan University, Shanghai 200040.
E-mail: hanssa_04@hotmail.com

Dr Li-Li Tao: Department of Pulmonary Medicine, Huadong Hospital, Shanghai Medical College, Fudan University, Shanghai 200040.
E-mail: ivydoctao@gmail.com

Dr Ying-Gang Zhu: Department of Pulmonary Medicine, Huadong Hospital, Shanghai Medical College, Fudan University, Shanghai 200040.
E-mail: robinzyg-zhu@vip.sina.com

Dr Hong-Ni Jiang: Department of Pulmonary Medicine, Zhongshan Hospital, Shanghai Medical College, Fudan University, Shanghai 200032.
E-mail: jhn2001@sina.com